Author's response to reviews

Title: Effect of resveratrol on alcohol induced mortality and liver lesions in mice

Authors:

Luis Bujanda (castro@medynet.com)
Maria Garcia-Barcina (castro@medynet.com)
Virginia Gutierrez-de Juan (castro@medynet.com)
Joseba Vidaurazaga (castro@medynet.com)
Marian Fernandez de Luco (castro@medynet.com)
Marian Gutierrez-Stampa (castro@medynet.com)
Mikel Larzabal (castro@medynet.com)
Elizabeth Hijona (castro@medynet.com)
Cristina Sarasqueta (castro@medynet.com)
Miguel Echenique-Elizondo (castro@medynet.com)
Juan I. Arenas (castro@medynet.com)

Version: 4 Date: 14 November 2006

Author's response to reviews: see over
Dear Sir,

Authors: Luis Bujanda, Virginia Gutierrez-deJuan, Joseba Vidaurazaga, Marian Fernandez de Luco, Marian Gutierrez-Stampa, Miguel Echenique-Elizondo, Mikel Larzabal, Elizabeth Hijona, Juan I. Arenas, Maria Garcia-Barcina and Cristina Sarasqueta

Title: Effect of resveratrol on alcohol induced mortality and liver lesions in mice

Journal: BMC Gastroenterology

MS: 8069038251117846

Please find enclosed the revised manuscript of “Effect of resveratrol on alcohol-induced mortality and liver lesions in mice,” which has been corrected in accordance with the “Core pre-accept”.

**Formatting changes requested**

1. Has been removed of the text: Title, Title page and Authors from the title page.
2. Has been removed the running foot from the title page.
3. Has been removed the full stop from the end of the title.
4. The authors have been written identically and in the order in which they are listed in the submission system. Superscript numbers have been used. Capitals have been removed.
5. Capitals has been removed of the correspondence.
6. Presentation details have been moved from title page to the Acknowledgements sections.
7. Has been used sentence case for all headings and sub-headings in the manuscript.
8. The abstract in the submission system is equal that in the manuscript file.
9. Keywords has been removed from the manuscript.
10. Introduction has been changed by Background.
11. References. All authors have been added to the references 3, 10, 19, 35 and 42.
12. Reference number 15 is mentioned in the text. The rest of references are correct in the text.
13. Material and methods has been changed by methods.
14. The last paragraph has been including in the conclusion section along with current sentence in “conclusions”
15. Acknowledgements. Is listed the source of funding for the study and the role of
the authors.
16. The title of Figure 1 is suppressed.
17. The white space around the image has been diminished.
18. A single composite illustration has been submitted of the figure 5.

We would be please to provide any additional information or modifications you may
require.

Yours sincerely,

Luis Bujanda